Author's response to reviews

Title: Elastography in predicting preterm delivery in asymptomatic, low-risk women: a prospective observational study.

Authors:

- Slawomir WOZNIAK (slavwo7572@gmail.com)
- Piotr CZUCZWAR (czuczwar@wp.pl)
- Piotr SZKODZIAK (piotr.szkodziak@gmail.com)
- Pawel MILART (pamilart@gmail.com)
- Ewa WOZNIAKOWSKA (ewa.wozniakowska@gmail.com)
- Tomasz PASZKOWSKI (tompasz@tlen.pl)

Version: 5 Date: 13 June 2014

Author's response to reviews: see over
Dear editorial board of BMC Pregnancy and Childbirth.

Please find enclosed the revised manuscript: "Elastography in predicting preterm delivery in asymptomatic, low-risk women: a prospective observational study." by Slawomir Wozniak et al. We are very grateful for the review. We made all the changes suggested by the editorial office. Please find below our answers to all the comments.

Sincerely yours,

Piotr Czuczwar and Slawomir Wozniak on behalf of the authors.

Corresponding author: Piotr Czuczwar at Medical University of Lublin, 3rd Chair and Department of Gynecology, 20-954 Lublin, ul. Jaczewskiego 8, Poland, czuczwar@wp.pl, phone number: +48 81 7244848

Revision response:

Editorial Comments:

1. Title Page

Please remove all author suffices from the title page. Please also copyedit Author surnames so that only the first letter is capitalized.

RESPONSE: The author list was modified as suggested.

2. Ethical Approval

Please provide the full name of the ethics committee which approved the
research. This should appear in your methods section.

**RESPONSE:** The full name of the ethics committee (Bioethical Committee of the Medical University of Lublin) was added in the M&Ms

3. Authors' contributions

Please confirm in this section that all authors read and approved the final manuscript. For formatting this section please see here:

http://www.biomedcentral.com/bmcpregnancychildbirth/authors/instructions/researcharticle#formatting-contributions

**RESPONSE:** The sentence “All authors read and approved the final manuscript.” was added in the authors’ contributions.